Author’s response to reviews

Title: Climate factors driven typhus group rickettsiosis incidence dynamics in Xishuangbanna Dai Autonomous Prefecture of Yunnan province in China, 2005-2017

Authors:
Yuan Gao (gaoyuancdc@126.com)
Yanlin Niu (niuyl@sina.com)
Wanwan Sun (sunwanwan321@126.com)
Keke Liu (kekegood@126.com)
Xiaobo Liu (liuxiaobo@icdc.cn)
Ning Zhao (zhaoning@icdc.cn)
Yujuan Yue (yueyujuan@icdc.cn)
Haixia Wu (wuhaixia@icdc.cn)
Fengxia Meng (mengfengxia@icdc.cn)
Jun Wang (wangjun@icdc.cn)
Xueshuang Wang (wangxueshuang@icdc.cn)
Qiyong Liu (liuqiyong@icdc.cn)

Version: 3 Date: 18 Dec 2019

Author’s response to reviews:

ENHE-D-19-00097R1

Climate factors driven typhus group rickettsiosis incidence dynamics in Xishuangbanna Dai Autonomous Prefecture of Yunnan province in China, 2005-2017

Yuan Gao; Yanlin Niu; Wanwan Sun; Keke Liu; Xiaobo Liu; Ning Zhao; Yujuan Yue; Haixia Wu; Fengxia Meng; Jun Wang; Xueshuang Wang; Qiyong Liu

Environmental Health
Dear Jan Semenza and Philippe Grandjean,

Thank you so much for your help! We have revised our manuscript and made some revisions. Looking forward to your letter! Thank you!

1. We have added the Consent for Publication in Declarations Section, as the following “This study has not been published previously or under consideration for publication elsewhere. The authors consent to publish the study in Environmental Health”.

2. We have corrected the abbreviations format as other papers do.

3. We have mentioned the comment 6 aspect in the limitation/discussion section.

4. We have added two funding sources in the funding section “the National Major Research and Development Program (grant number 2016YFC1200802), and the China Prosperity Strategic Programme Fund (SPF) 2015-16 (Project Code: 15LCI1)”. 

Best regards,

Yuan Gao